Author's response to reviews

Title: Creating a medical dictionary using word alignment: The influence of sources and resources

Authors:

Mikael Nystrom (mikny@imt.liu.se)
Magnus Merkel (magme@ida.liu.se)
Hakan Petersson (hakpe@imt.liu.se)
Hans Ahlfeldt (hanah@imt.liu.se)

Version: 4 Date: 23 November 2007

Author's response to reviews: see over
Dear Editor,

The manuscript has been updated according to the requests below.

- Table title and legend. The table title should appear above the table. The legend (or key) should appear below the table.

- Table shading. We are unable to display shading and colours within the tables. Can we therefore ask you to use another method of highlighting the shaded text e.g. by making it bold.

- Glossary- please include this as an appendix

- Appendices- please move these to before the acknowledgments.

Best regards,
Mikael Nyström